Author’s response to reviews

Title: Statin and the risk of hepatocellular carcinoma in patients with hepatitis B virus or hepatitis C virus infection: a meta-analysis

Authors:

Xiaofei Li (metshep@sina.com)
Lina Sheng (shenglina_8892@sohu.com)
Liwen Liu (liuliwen_8865a@tom.com)
Yongtao Hu (yongtao_hu89@sohu.com)
Yongxin Chen (chenyx_9882a@yeah.net)
Lianqing Lou (llq982_a@sohu.com)

Version: 2  Date: 26 Feb 2020

Author’s response to reviews:

Yasser Fouad (Reviewer 1):

Interesting study and well written. Other confounding factors should have been mentioned specifically as chronic alcoholism and metabolic liver diseases

Authors’ response: Thanks for your comments. We have mentioned chronic alcoholism and metabolic liver diseases as potential confounding factors in the limitation section of the revised manuscript (Page 15, Line 18).